Author's response to reviews

Title: An Asian viewpoint on the use of vitamin D and calcium in osteoporosis treatment: Physician and patient attitudes and beliefs

Authors:

Siew Pheng Chan (shuvayu_sen@merck.com)
Boyd B Scott (boyd_scott@merck.com)
Shuvayu S Sen (shuvayu_sen@merck.com)

Version: 2 Date: 15 January 2010

Author's response to reviews:

I have attached copies of the patient and physician questionnaires.

I have clarified in the METHODS the requirements for ethical approval of the study (bold text).

I have added a more detailed competing interest section to the manuscript (bold text).